Author's response to reviews

Title: Antioxidant activity and hepatoprotective potential of Cedrelopsis grevei on cypermethrin induced oxidative stress and liver damage in male mice

Authors:

Abdel-Tawab H Mossa (abdeltawab.mossa@yahoo.com)
Tarek Heikal (tarekhl@yahoo.com)
Meriam Belaiba (belaibameriam@gmail.com)
Emmanuel Raoelison (guyroael@gmail.com)
Hicham Ferhout (hferhout@natex-biotech.com)
Jalloul Bouajila (jalloul.bouajila@univ-tlse3.fr)

Version: 2
Date: 10 March 2015

Author's response to reviews: see over
Dear Editor, Journal of BMC Complementary and Alternative Medicine
We highly appreciate the detailed valuable comments of the referees on my manuscript. The suggestions are quite helpful for us and we incorporate them in the revised paper. You will note our changes/amendments are written in red color.. We hope the Reviewers and the Editors will be satisfied with our responses to the ‘comments’ and the revisions for the original manuscript. However, if you need any further inquiries, please inform me. Thanking you in advance for your kind co-operation. With kind regards,
Dr. Abdel-Tawab H. Mossa
Associate Professor
Environmental Toxicology Research Unit (ETRU), Pesticide Chemistry Department, National Research Centre, Tahrir Street, B.O.Box 12311, Dokki, Cairo, EGYPT.
Tel.: (202)-33371211/33371615; Fax: (202)-33370931
E-mail: abdeltawab.mossa@yahoo.com